Author's response to reviews

Title: Knowledge and Attitude towards Rape and Child Sexual Abuse - a community-based cross-sectional study in Rural Tanzania

Authors:

Muzdalifat S Abeid (muzsalim@yahoo.com)
Projestine S Muganyizi (promuga@yahoo.com)
Siriel N Massawe (snanzia@gmail.com)
Rose C Mpembeni (rcmpembeni@yahoo.com)
Elisabeth Darj (elisabeth.darj@kbh.uu.se)
Pia Axemo (pia.axemo@kbh.uu.se)

Version: 3 Date: 15 October 2014

Author's response to reviews: see over
15 October 2014

To the Editor-in-Chief

BMC Public Health

Dear Sir/Madam,

On behalf of my co-others, I am re-submitting a manuscript entitled “Knowledge and Attitude towards Rape and Child Sexual Abuse – a community-based cross-sectional study in Rural Tanzania” for possible publication in your esteemed journal BMC Public Health. We really appreciate the Editors comments and relevant suggestions. More information has been included under the results section and two figures to illustrate the gender differences in the responses. We have also included a table indicating the specific questions for knowledge and attitude. We hope this has improved our manuscript and provides clearer picture. The authors’ details have not changed.

This manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with submission to BMC Public Health. The study was supported by a grant from the Sida/SAREC. The authors have no conflicts of interest to declare.

We look forward to hearing from you at your earliest convenience.

Yours sincerely

Muzdalifat Abeid
muzsalim@yahoo.com